Author's response to reviews

Title: Cold Saline Irrigation of the Renal Pelvis During Radiofrequency Ablation of a Central Renal Neoplasm

Authors:

Jose Pablo Morales (jpmorales@doctors.org.uk)
Tarun Sabharwal (Tarun.Sabharwal@gstt.nhs.uk)
Marios Georganas (gmarios@doctors.org.uk)
Renato Dourado (Renato.Dourado@gstt.nhs.uk)
Declan Cahill (Declan.Cahill@gstt.nhs.uk)
Andreas Adam (Andy.Adam@kcl.ac.uk)

Version: 7 Date: 5 July 2007

Author's response to reviews: see over
In summary, we respectfully suggest that Dr Wah has not met her responsibilities as a reviewer because she has made unsubstantiated statements and has not explained her reasoning. Reviewer One, on the other hand felt very positive about our report and felt that it would make a difference to clinical practice and had encouraged publication. Therefore, we suggest that a third reviewer is asked to look at this manuscript and to address the points we have made.

Thank you very much for your consideration.

Yours sincerely

A Adam
Professor of Interventional Radiology

T Sabharwal
Consultant Interventional Radiologist
